Author's response to reviews

Title: Socio-demographic factors related to periodontal status and tooth loss of pregnant women in Mbale district, Uganda.

Authors:

Margaret Wandera (Margaret.Wandera@odfa.uib.no)
Ingunn MS Engebretsen (Ingunn.Engebretsen@cih.uib.no)
Isaac Okullo (iokullo@med.mak.ac.ug)
James K Tumwine (jtumwine@imul.com)
Anne N Åstrom (Anne.Astrom@iko.uib.no)
PROMISE-EBF (PROMISE-EBF) (Anne.Astrom@iko.uib.no)

Version: 7 Date: 10 July 2009

Author's response to reviews: see over
Submission of revised manuscript

We are delighted to hear our manuscript will be published in your prestigious journal. We have made the editorial changes to the author list and acknowledgements as requested and now submit a final version of “Socio-demographic factors related to periodontal status and tooth loss of pregnant women in Mbale district, Uganda”.

Thank you for your positive consideration.

Yours sincerely

Margaret Wandera.
<table>
<thead>
<tr>
<th></th>
<th><strong>Major revisions</strong> (We require the author to make these changes)</th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>Author list - The list of authors in the manuscript should be exactly as they are in the submission system.</td>
<td>• We have matched the author list on manuscript with submission in the system and removed duplicate names from author group in acknowledgements.</td>
</tr>
<tr>
<td></td>
<td>If the authors in the author group are listed in the acknowledgements they should be removed from the author list. Authors not included in the author group should appear in the author list both on the title page of your manuscript and in the submission system</td>
<td>• We are submitting latest version.</td>
</tr>
<tr>
<td></td>
<td>Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.</td>
<td></td>
</tr>
<tr>
<td></td>
<td>Please make sure you use the latest version of the main file as we have made several formatting changes and using a previous unformatted version will delay publication.</td>
<td></td>
</tr>
</tbody>
</table>